Author's response to reviews

Title: Eye exercises: their impact on myopia and ocular fatigue in Chinese urban students

Authors:

Zhong Lin (linzhong_@126.com)
Balamurali Vasudevan (darmurali@gmail.com)
Vishal Jhanji (vishaljhanji@gmail.com)
Tieying Gao (gaotieying520@yahoo.com.cn)
Ningli Wang (linzhong85@gmail.com)
Qi Wang (linzhong85@foxmail.com)
Ji Wang (zhoujia115618@126.com)
Kenneth J Ciuffreda (kciuffreda@sunyopt.edu)
Yuanbo Liang (bmps.nitm@gmail.com)

Version: 2 Date: 29 January 2013

Author's response to reviews:

Dear Editor(s):

We have added the Competing interests, Financial competing interests, Authors’ contributions, and Acknowledgements in our manuscript (page 10) after Conclusions section as required.

Thanks very much!